Author's response to reviews

Title: Evolving epidemiology and antimicrobial resistance in spontaneous bacterial peritonitis: a two-year observational study

Authors:

lionel Piroth (lionel.piroth@chu-dijon.fr)
andré Péchinot (andre.pechinot@chu-dijon.fr)
v Vincent Di Martino (v.dimartino@chu-besancon.fr)
Yves Hansmann (Yves.Hansmann@chru-strasbourg.fr)
Alain Putot (alain.putot@chu-dijon.fr)
Isabelle Patry (ipatry@chu-besancon.fr)
Tahar Hadou (t.hadou@chu-nancy.fr)
Benoit Jaulhac (jaulhac@unistra.fr)
Catherine Chirouze (cchirouze@chu-besancon.fr)
Christian Rabaud (c.rabaud@chu-nancy.fr)
Alain Lozniewski (a.lozniewski@chu-nancy.fr)
Catherine Neuwirth (catherine.neuwirth@chu-dijon.fr)
Pascal Chavanet (pascal.chavanet@chu-dijon.fr)
Anne Minello (anne.minello@chu-dijon.fr)

Version: 5
Date: 12 May 2014

Author's response to reviews: see over
1) I deeply apologize since the reference referred to what was used in our previous study. The link toward updated CASFM recommendations has been added in the text and the previous reference deleted.

2) We indeed used the same definitions than that proposed by the international panel. I have added and distinguished the main antibiotic classes routinely tested in the different labs of the study group, which should help in better understanding how the bacteria were classified.

Thank you for your support and your explanations.

I hope that it will now meet your approval and I apologize for this delay.